

Citation: Chiavenna C, Presanis AM, Charlett A, de Lusignan S, Ladhani S, Pebody RG, et al. (2019) Estimating age-stratified influenza-associated invasive pneumococcal disease in England: A timeseries model based on population surveillance data. PLoS Med 16(6): e1002829. https://doi.org/ 10.1371/journal.pmed.1002829

Academic Editor: Aziz Sheikh, Edinburgh University, UNITED KINGDOM

Received: November 19, 2018

Accepted: May 17, 2019

Published: June 27, 2019

Copyright: © 2019 Chiavenna et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data underlying the results presented in the study are owned by third parties and authors do not have permission to share them. Data are freely available upon request from Ana Correa at the Royal College of General Practitioners, for the GP consultations data (email a.correa@surrey.ac.uk); Praveen SebastianPillai at Public Health England, respiratory department, for the virological positivity data (email <u>Praveen</u>. <u>Sebastianpillai@phe.gov.uk</u>); Zahin Amin at Public Health England, immunisation department, for the RESEARCH ARTICLE

Estimating age-stratified influenza-associated invasive pneumococcal disease in England: A time-series model based on population surveillance data

Chiara Chiavenna^{1*}, Anne M. Presanis¹, Andre Charlett², Simon de Lusignan³, Shamez Ladhani², Richard G. Pebody², Daniela De Angelis^{1,2*}

MRC Biostatistics Unit, School of Clinical Medicine, University of Cambridge, Cambridge, United Kingdom,
 National Infection Service, Public Health England, London, United Kingdom,
 Department of Clinical and Experimental Medicine, University of Surrey, Surrey, United Kingdom

* chiara.chiavenna@mrc-bsu.cam.ac.uk

Abstract

Background

Measures of the contribution of influenza to *Streptococcus pneumoniae* infections, both in the seasonal and pandemic setting, are needed to predict the burden of secondary bacterial infections in future pandemics to inform stockpiling. The magnitude of the interaction between these two pathogens has been difficult to quantify because both infections are mainly clinically diagnosed based on signs and symptoms; a combined viral–bacterial testing is rarely performed in routine clinical practice; and surveillance data suffer from confounding problems common to all ecological studies. We proposed a novel multivariate model for age-stratified disease incidence, incorporating contact patterns and estimating disease transmission within and across groups.

Methods and findings

We used surveillance data from England over the years 2009 to 2017. Influenza infections were identified through the virological testing of samples taken from patients diagnosed with influenza-like illness (ILI) within the sentinel scheme run by the Royal College of General Practitioners (RCGP). Invasive pneumococcal disease (IPD) cases were routinely reported to Public Health England (PHE) by all the microbiology laboratories included in the national surveillance system. IPD counts at week *t*, conditional on the previous time point *t*–1, were assumed to be negative binomially distributed. Influenza counts were linearly included in the model for the mean IPD counts along with an endemic component describing some seasonal background and an autoregressive component mimicking pneumococcal transmission. Using age-specific counts, Akaike information criterion (AIC)-based model selection suggested that the best fit was obtained when the endemic component was expressed as a function of observed temperature and rainfall. Pneumococcal transmission within the same age group was estimated to explain 33.0% (confidence interval [CI] 24.9%–39.9%) of new

invasive pneumococcal disease data (email <u>zahin</u>. amin@phe.gov.uk).

PLOS MEDICINE

Funding: CC was supported by a Public Health England studentship and funds from the NIHR Health Protection Unit on Evaluation of Interventions. AMP and DDA were supported by the Medical Research Council (Unit programme number MC_UU_00002/11). No funding bodies had any role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: SdL, as part of his academic role, sits on Advisory Boards for Sanofi and Seqirus, and he received research funding through University of Surrey from GSK and Seqirus.

Abbreviations: AIC, Akaike information criterion; ARI, acute respiratory illness; ARIMA, autoregressive integrated moving average; CI, confidence interval; GATHER, Guidelines for Accurate and Transparent Health Estimates Reporting; GLM, generalized linear model; hMPV, human metapneumovirus; ILI, influenza-like illness; IPD, invasive pneumococcal disease; LRTI, lower respiratory tract infection; PHE, Public Health England; RCGP RSC, Royal College of General Practitioners Research and Surveillance Centre; RT-PCR, reverse transcription PCR; RSV, respiratory syncytial virus; SGSS, Second Generation Surveillance System. cases in the elderly, whereas 50.7% (CI 38.8%–63.2%) of incidence in adults aged 15–44 years was attributed to transmission from another age group. The contribution of influenza on IPD during the 2009 pandemic also appeared to vary greatly across subgroups, being highest in school-age children and adults (18.3%, CI 9.4%–28.2%, and 6.07%, CI 2.83%–9.76%, respectively). Other viral infections, such as respiratory syncytial virus (RSV) and rhinovirus, also seemed to have an impact on IPD: RSV contributed 1.87% (CI 0.89%–3.08%) to pneumococcal infections in the 65+ group, whereas 2.14% (CI 0.87%–3.57%) of cases in the group of 45- to 64-year-olds were attributed to rhinovirus. The validity of this modelling strategy relies on the assumption that viral surveillance adequately represents the true incidence of influenza in the population, whereas the small numbers of IPD cases observed in the younger age groups led to significant uncertainty around some parameter estimates.

Conclusions

Our estimates suggested that a pandemic wave of influenza A/H1N1 with comparable severity to the 2009 pandemic could have a modest impact on school-age children and adults in terms of IPD and a small to negligible impact on infants and the elderly. The seasonal impact of other viruses such as RSV and rhinovirus was instead more important in the older population groups.

Author summary

Why was this study done?

- Since the deadly 1918 Spanish flu, pandemic preparedness has been crucial for many governments: in the United Kingdom, a pandemic is high on the government's national risk register of civil emergencies [1].
- Previous research has shed light on the central role of secondary bacterial infections, suggesting a synergistic interplay of influenza virus and *S. pneumoniae*, a bacterium that infects the lungs.
- Quantifying the magnitude of such interaction at the population level is of central importance to inform public health policy: in England, current decision-making on required sizes of an antibiotic stockpile for use in a future pandemic lacks scientific evidence.

What did the researchers do and find?

- We analysed routinely collected population-level counts of invasive pneumococcal disease (IPD) cases.
- We decomposed the incidence of IPD in terms of endemic pneumococcal circulation, a seasonal contribution, and an influenza-driven component.

 We proposed a novel multivariate age-stratified modelling framework to assess the contribution of influenza across seasons and age groups.

What do these findings mean?

- We found that influenza played a more important role towards explaining IPD during the 2009 pandemic than during seasonal epidemics.
- This role was particularly prominent in school-age children.
- These results are valuable to quantify the possible contribution of influenza to the burden of IPD in a future pandemic of influenza with similar characteristics to the 2009 pandemic.

Introduction

Just one century ago, the "1918 Spanish Influenza" is thought to have caused at least 50 million deaths worldwide despite influenza often naively being considered to be a nonsevere disease. Hence, a number of researchers in recent decades have tried to understand the drivers of such severity in the fear of a new pandemic [2–4]. Viral–bacterial synergism, in particular with *S. pneumoniae*, is considered to have played a major role in the observed mortality rate, as postmortem examinations revealed the presence of bacteria in the lungs of many influenza-infected individuals [5].

The synergistic interplay between influenza and *S. pneumoniae* has been validated in animal models [6]; however, routine ascertainment of coinfection remains difficult and expensive in humans [7]: individual-level data on the exposure are hard to acquire because pathogens often circulate silently within a host population or manifest themselves through nonspecific clinical symptoms [8–10]. Infections due to each pathogen are separately identified in the presence of a corresponding disease, and the likelihood of an improved understanding of pathogen inter-actions strongly relies on indirect inference.

Time series of respiratory diseases are characterised by strong seasonal patterns, with an increased incidence in the winter months in temperate areas of the world. Disentangling the contribution towards *S. pneumoniae* of the influenza virus from other risk factors that exhibit the same seasonal variation (e.g., weather, daylight, circulation of other pathogens, etc.) [11] can be challenging. A variety of regression methods have been suggested in the general framework of burden estimation, especially for excess morbidity and mortality due to seasonal and pandemic influenza [12, 13].

Using sentinel testing of suspected influenza cases, the presence and magnitude of influenza virus in the community is usually summarised by the proportion of positive tests by viral type (and/or subtype). The so-called virological regression model includes influenza circulation as a covariate in a cyclic regression model for respiratory infections, in which seasonality of disease is described by sine and cosine terms [14, 15]. Lagged effects of virological circulation, or other confounders that exhibit annual variation in intensity or timing, can also be included [16]. The most adequate distribution for the outcome variable has been widely debated: [17] argued that counts of disease should be modelled as Poisson distributed, employing a log-link

function; however, such a link implies an exponential increase of the outcome with respect to the number of confirmed influenza cases and multiplicative effects of covariates (i.e., respiratory viruses). As these assumptions are quite unrealistic, [18] and [19] suggested the use of a generalized linear model (GLM) with a Poisson error distribution but identity link [20, 21].

Previous work estimated the burden of influenza on syndromic healthcare contacts, such as lower respiratory tract infection (LRTI) [3], acute respiratory illness (ARI) [22], or respiratory hospital admissions [23]; however, this has not elucidated the relative contribution of the interaction between influenza virus and *S. pneumoniae* relative to shared seasonality [3, 24, 25]. Reference [26] estimated the percentage of invasive pneumococcal disease (IPD) cases attributable to influenza and respiratory syncytial virus (RSV) using regression models; however, since we are dealing with a transmissible pathogen, the independence among observations they assume is unlikely to hold. Autoregressive integrated moving average (ARIMA) models have also been proposed [27]; however, such an approach and its multivariate counterpart, ARIMAX, require applying preliminary transformations to the original data when non-stationary behaviour is detected. The necessity of choosing model order via an empirical procedure based on model fit, along with the limited interpretability of coefficients, precludes ARIMA methods as a sensible choice for our scope [28].

We combined the key strengths of each of the previous approaches into a single flexible regression model, following the work of [29]: weekly IPD counts were decomposed into an endemic component, with sine-cosine waves describing cyclic winter outbreaks, and an epidemic autoregressive component, in which lagged IPD counts entered the model linearly using an identity link function [30]. A time-varying covariate could also be linearly added to the model, with the corresponding coefficient expressing the association between the two time series after taking into account shared drivers [31].

We extended the modelling framework of [29] to address a number of issues. First, we were interested in investigating the contribution of several pathogens to the incidence of IPD: other viruses such as RSV and rhinovirus have been speculated to interact with S. pnuemoniae, showing an association with an increased risk of IPD [32, 33]. In contrast to this existing work, we jointly modelled the epidemic evolution of viruses of interest by simultaneously including them as covariates in models of the IPD counts. Secondly, associations between pathogens have been suggested to be heterogeneous across age groups [34]; hence, we implemented multivariate versions of the model allowing estimation of age-specific associations. The multivariate structure also permitted decomposition of IPD transmission between and across age groups by incorporating contact patterns. Finally, as there is evidence that meteorological conditions such as temperature and humidity affect seasonality and intensity of outbreaks [35, 36], we replaced sinusoidal functions with observed weather information. Compared to previous work [37, 38], we proposed a phenomenological model that expresses IPD dynamics as a function of autoregressive components, viral infections, age-specific contact patterns, and seasonal confounders without making strong assumptions on the transmission mechanism, aiming to provide a parsimonious characterisation of the drivers of IPD patterns over time.

Methods

Data

Influenza is generally diagnosed based on influenza-like illness (ILI), defined as the simultaneous presence of signs and symptoms such as high fever, cough, and myalgia; however, only virological testing allows the ascertainment of the responsible pathogen. For this reason, we estimated influenza incidence by combining two data sources. The Royal College of General Practitioners Research and Surveillance Centre (RCGP RSC) collects weekly numbers of general practice consultations for several clinical diagnoses of communicable and respiratory diseases, including ILI. The population monitored by the RCGP RSC practices covers an average population of approximately 1.4 million persons, 2.6% of England, considered to be representative of the national population in terms of age, gender, deprivation index, and prescription patterns [39]. As part of routine virological surveillance, in general practices participating in the RCGP RSC scheme, a proportion of ILI cases is swabbed and the samples are tested for influenza A (H1 or H3 subtypes), influenza B, RSV, and human metapneumovirus (hMPV) by the Public Health England (PHE) reference laboratory [39]. The number of specimens tested and the number of positives for each virus were stratified by week of test and age group to derive the proportion of virologically positive specimens. This proportion was then multiplied by ILI counts to compute the corresponding age and time-specific consultations attributable to influenza.

S. pneumoniae (the pneumococcus) infection is often asymptomatic, as this is a commensal bacterium of the human nasopharynx; nonetheless, its progression to the lower respiratory tract and blood can cause severe disease, namely IPD. In the UK, counts of positive isolates for a number of clinically significant pathogens are reported weekly to PHE by all the microbiology laboratories included in the national surveillance system and are stored in the Second Generation Surveillance System (SGSS) database. Counts of IPD, RSV, and rhinovirus infections were extracted from SGSS. Consistency in testing over time and space was guaranteed by the 'United Kingdom Standards for Microbiology Investigations', a diagnostic algorithm applied across laboratories to patients presenting with different clinical syndromes [40]. Finally, estimates of the population of England by age group, during each season, were obtained from the Office for National Statistics [41], and weather information such as daily central England temperature and daily England and Wales precipitation were downloaded from the MetOffice HadCET data repository [42].

This study is reported as per the Guidelines for Accurate and Transparent Health Estimates Reporting (GATHER) [43] (S1 Checklist). The study did not have a prospective design, and analysis was planned as we retrospectively gathered information on routinely collected and publicly available data sources, considering England as being representative of temperate areas in the northern hemisphere, where the time series of interest feature typical winter peaks. The time period considered ranged from 1 January 2009 to 31 December 2017, with the 2009 pandemic period defined to include the three waves, from week 15/2009 to week 26/2011 [44]. Disease incidence was categorised into five age groups: 0–4, 5–14, 15–44, 45–64, and 65+ years old, as in similar studies [26].

Statistical model

When dealing with two strongly associated time-series representing infectious disease incidence, the modelling framework presented by [31] allows quantification of the relationship between outbreaks of the two pathogens of interest. Denoting by Y_t the random variables representing counts of disease Y at weeks t = 1, ..., T, it is assumed they are Poisson distributed: $Y_t | Y_{t-1} \sim Poi(\mu_t)$, with conditional mean μ_t expressed as

$$\mu_t = pop_t v_t + \lambda Y_{t-1} + \tau X_{t-1}, \tag{1}$$

where v_t is an endemic log-linear predictor that, multiplied by an offset such as population size pop_t , might describe incidence due to seasonal or sociodemographic variation; Y_{t-1} is a temporal interaction (epidemic component) whose coefficient λ represents the transmission of infection from time t-1 to time t; and X_{t-1} are lagged counts of disease X, with the coefficient τ quantifying the strength of association between Y_t and X_{t-1} . For overdispersed counts, the

Poisson distribution for the observation model can be replaced by a negative binomial with overdispersion parameter ψ :

$$Y_t | Y_{t-1} \sim NegBin(\mu_t, \psi).$$
⁽²⁾

The decomposition of the contribution of several phenomena in additive components, along with the small number of parameters, makes interpretation very straightforward while preserving biologically meaningful relationships among the quantities of interest. Moreover, compared to the parameter-driven models briefly reviewed in the introduction that are characterised by harmonic functions, the presence of an observation-driven component in model (1) could capture outbreaks more easily, as λ expresses the additional temporal dependence beyond the seasonality explained by the parametric model [45]. Finally, modelling overdispersion instead of assuming Poisson-distributed outcomes allows further flexibility.

The model in Eq 1 can be easily extended to deal with stratified time series: [46] implemented a multivariate version for spatial disease spread and later embedded an age-structure into it [47]. We followed a similar approach when modelling disease counts in age group a, $Y_{a,t}$, where $a \in \{0-4, 5-14, 15-44, 45-64, 65+\}$. Two transmission components were included at this stage:

$$\mu_{t,a} = pop_{t,a}v_{t,a} + \lambda_a Y_{t-1,a} + \phi_a \sum_{k \neq a} c_{k,a} Y_{t-1,k \neq a} + \tau_a X_{t-1,a}.$$
(3)

In addition to the transmission of one pathogen within age group a, quantified by λ_a , we explicitly incorporated the transmission of the same pathogen across age groups through ϕ_a . In order to account for heterogeneity of contact patterns, counts of disease in groups $k \neq a$ were weighted by the element $c_{k,a}$ of a contact matrix (e.g., POLYMOD or any other measure of social distancing between group k and a [48]). Hence, the coefficient ϕ_a , paired with such a linear combination of disease cases, represents the contribution of transmission from other population subgroups to disease in age group a. Both transmission coefficients were specified to be age-specific because, despite accounting for contact patterns, some age groups are known to be more susceptible to infection than others. We also allowed heterogeneity across groups for the remaining model parameters, as the interaction between influenza and *S. pneumoniae* has also been suggested to vary with age [7]. Finally, we extended this setting to incorporate more than two time series, estimating the association of the outcome of interest with more than one pathogen (e.g., other indicators of viral circulation such as rhinovirus and RSV incidence).

Models in Eqs 1 and 2 are both implemented in the R package 'surveillance' through the hhh4 function. For the model in Eq 3, in which multiple covariates were added, our algorithm simultaneously fitted models for different strata incorporating the contact structure. Similarly to the hhh4 function, we also obtained maximum-likelihood estimates via a (globally convergent) Newton-Raphson type algorithm. To ensure positivity, parameters were optimised on the log-scale—i.e., $log(\psi)$ and $log(\lambda)$ were used. Uncertainty about the proportions of IPD cases attributable to each virus was estimated by resampling n = 10,000 datasets from the fitted model and taking the 95% confidence intervals (CIs) to be the empirical 2.5% and 97.5% percentiles across the resampled datasets.

Results

A total of 62,679 ILI consultations within the sentinel scheme and of 45,601 IPD cases nationwide have been notified over 9 years. Fig 1 displays the temporal trend of all ILI and influenzaconfirmed consultation rates respectively, where influenza-confirmed counts (referred to as





'Flu' from now on) were obtained as described in the Methods. A clear seasonal pattern is visible, with regular outbreaks in the winter months and epidemics lasting 10–15 weeks, except for 2009, when the A/H1N1 pandemic started in spring. Virological testing is not systematically performed during the summer; hence, the Flu data are quite sparse off-season. Nonetheless, it is evident how, even during winter, the influenza cases do not closely mimic the ILI curve, confirming the nonspecificity of the ILI diagnosis. In the IPD time series (Fig 1, bottom panel), peaks appear to be similar across seasons both in terms of amplitude and timing, with a gradual increase of cases from autumn to a winter peak, followed by a decline in summer. The incidence rate per 1,000,000 population is plotted in this case, as IPD is rare.

We followed the analysis strategy reported in full in the <u>S1 Appendix</u>. Briefly, the best formulation for the model in Eq 1 was first identified in terms of Akaike information criterion (AIC) values. A summary of model comparison is presented in <u>Table 1</u>: starting from a Poisson distributional assumption and one set of harmonic functions (S = 1, see <u>S1 Appendix</u>), more

	Distribution	Endemic	Covariate	AIC	log(s[P,x])
A	Poi	S = 1	Flu	5,107.61	5.805
В	NB	S = 1	Flu	4,043.95	4.408
С	NB	rain + temp, lag = 1	Flu	4,029.19	4.400
D	NB	rain + temp, lags = 5 ($p_{weather} = 0.8$)	Flu	4,027.82	4.390
E	NB	rain + temp, lags = 5 ($p_{weather} = 0.8$)	Flu + rhinov	3,997.93	4.361
F	NB	rain + temp, lags = 5 ($p_{weather} = 0.8$)	Flu + rhinov + RSV	3,992.95	4.334

Table 1. Model comparison in terms of AIC and one-step-anead forecast (logis(P,X))	Table 1.	Model comparison	in terms of AIC a	and one-step-ahead	forecast (log[s(P,x)]).
--	----------	------------------	-------------------	--------------------	-------------------------

Abbreviations: AIC, Akaike information criterion; NB, negative binomial; Poi, Poisson; rhinov, rhinovirus; RSV, respiratory syncytial virus; temp, temperature

https://doi.org/10.1371/journal.pmed.1002829.t001

complicated versions of the endemic component were assessed by replacing trigonometric waves with weather variables (model C). We also tested whether multiple lags for covariates better described the observed patterns: considering lags q = 1, ..., Q where Q = 5—i.e., including up to 5 weeks before time t, we saw no gain in adding either Flu or IPD lagged counts when q>1. The only variables whose lagged values improved model fit were rainfall and temperature; nonetheless, the parameter representing the decline in weight attributed to lagged values was optimally chosen to be $p_{weather} = 0.8$, suggesting that only 20% of the weight is attributed to observations more than one week before (model D).

Evaluating the model in terms of one-step-ahead forecasts, we selected 30 weeks as the initial time window of observed data, and we repeated the forecast for each of the remaining 440 weeks. We reassuringly found mean log(s(P,x)) (see S1 Appendix) to be minimal for the endemic formulation including weather information, with lags weighted according to $p_{weather} = 0.8$ (model D).

Fitted values for all components according to model formulations B and D are shown in Figs 2 and 3. The number of IPD cases attributed to Flu during the entire study period was as







low as 199 according to model D, including weather variables—i.e., 0.45% (CI < 0.01%– 1.59%) of all the IPD cases. However, 100 of these cases happened during the three pandemic waves, 0.83%, CI < 0.01%–2.94%, of all the observed IPD cases in that period, suggesting that the pandemic strain might have been responsible for an increased incidence. As a sensitivity analysis, we selected Flu counts referring only to the three pandemic waves: the increase in AIC was minimal compared to model D including Flu counts over all the study period, suggesting that the role of seasonal Flu is marginal. We also considered each season as a separate covariate, with results plotted in S1 Fig.

Finally, we investigated whether other viruses also interact with *S. pneumoniae*: the number of rhinovirus (model E in Table 1) and RSV (model F) infections were sequentially added to the selected model D (the observed time series are plotted in S2 and S3 Figs). Rhinovirus alone greatly enhanced the fit to the data, and the inclusion of RSV on top of Flu and rhinovirus still resulted in model improvement. Hence, the best-fitting model (F) for mean IPD counts at time *t* takes the form

$$u_{IPD,t} = pop_{t}[exp(\alpha + \gamma \sum_{q=1}^{5} w_{q}(weather)temp_{t-q} + \delta \sum_{q=1}^{5} w_{q}(weather)rain_{t-q})] + \lambda IPD_{t-1} + \tau Flu_{t-1} + \theta rhinovirus_{t-1} + \zeta RSV_{t-1}$$

$$(4)$$

with overdispersion parameter ψ and decay parameter for w_q (weather) fixed to $p_{weather} = 0.8$. Point estimates and standard errors for the coefficients are reported in Table 2, and relative contributions are pictured in Fig 4: rhinovirus explained 6.97% (CI 4.27%–10.28%) of all the IPD cases, 2.48% (CI 0.51%–4.52%) were attributed to RSV, and only 0.67% (CI < 0.01%–1.69%) were attributed to Flu. Overall, the three viruses accounted for 10.12% (CI 7.18%–13.77%) of IPD cases at population level.

Selected plots displaying age-specific incidence can be found in $\underline{S4}-\underline{S6}$ Figs. For consistency, we used for all age groups the distributional assumption and the endemic component that

	Estimate	Std. Error	
α	-3.1831	0.0066	
γ	-0.4572	0.0052	
δ	-0.0838	0.0034	
$log(\psi)$	3.5064	0.0080	
$log(\tau)$	-1.2946	0.3240	
log(θ)	2.9821	0.0193	
log(ζ)	1.8944	0.0742	
$log(\lambda)$	5.8521	2e-04	

Table 2. Coefficient estimates for the model of IPD including Flu, rhinovirus, and RSV as covariates.

Abbreviations: IPD, invasive pneumococcal disease; RSV, respiratory syncytial virus; Std., standard

https://doi.org/10.1371/journal.pmed.1002829.t002

fitted the univariate time series best (model D). Thus, when considering attribution of IPD to Flu, model selection started by considering the model in Eq 3: $IPD_{t,a}|IPD_{t-1,a} \sim NegBin(\mu_{IPD,t,a}, \psi_a)$ where

$$\mu_{IPD,t,a} = pop_{t,a}[exp(\alpha_a + \gamma_a \sum_{q=1}^{5} w_q(weather)temp_{t-q} + \delta_a \sum_{q=1}^{5} w_q(weather)rain_{t-q})] + \tau_a FluA_{t-1,a} + \lambda_a IPD_{t-1,a} + \phi_a \sum_{k \neq a} c_{k,a} IPD_{t-1,k \neq a}$$
(5)

However, this required estimating 35 coefficients, not a very parsimonious option. Hence, we tried model reduction by testing whether any of the coefficients could be the same across groups. Full model comparison is reported in <u>S1 Table</u>. AIC decreased from 13,218.85 (model G, with all age-specific coefficients) to 13,216.32 by using a shared rainfall coefficient, i.e., $\delta_a =$



Fig 4. Model (F) including influenza, rhinovirus, and RSV. IPD, invasive pneumococcal disease; RSV, respiratory syncytial virus.

Age, years	Pneumococcal Transmission		Influenza A	
	Within Group	Across Groups	Overall	AH1N1pmd09
<5	26.42 (16.16-34.49)	18.73 (4.67–33.14)	0.00	0.00
5-14	15.70 (5.32–24.17)	27.32 (2.18-55.97)	8.40 (4.12-13.66)	18.30 (9.43–28.16)
15-44	19.47 (10.33–27.07)	50.67 (38.79-63.16)	3.55 (1.64-5.76)	6.07 (2.83-9.76)
45-64	23.65 (14.79-31.03)	41.64 (31.61-51.49)	0.92 (<0.01-2.94)	1.19 (<0.01-3.78)
65+	33.02 (24.89–39.88)	34.45 (26.02-43.24)	0.00	0.00

Table 3. Model I: Relative proportions (%) of IPD cases attributed to pneumococcal transmission within and across age groups and to influenza overall or in the pandemic period.

Abbreviations: IPD, invasive pneumococcal disease. https://doi.org/10.1371/journal.pmed.1002829.t003

 δ for any age (model H). Finally, the utility of multiple lags for Flu and IPD was considered, but once again a benefit from including past values only pertained to weather variables.

Estimated coefficients and standard errors for model H are shown in S2 and S3 Tables. The τ_a parameters associated with influenza were quite heterogeneous across age groups, showing an inverse U–shaped tendency: almost null in young children and the elderly and more prominent in other age groups. However, because of the very small size and associated large uncertainty of the parameters τ_5 and τ_{65+} , we refitted the model fixing them to zero (model I). The attributed proportions of IPD cases estimated from this model are reported in Table 3, estimated coefficients and standard errors are shown in Tables 4 and 5, and fitted values for all age groups are plotted in Figs 5–9.

According to model I, IPD was driven by Flu in school-age children (8.40%, CI 4.12%–13.66%) and adults aged 15–44 years (3.55%, CI 1.64%–5.76%), and these components were strikingly higher in the pandemic period: 18.30% (CI 9.43%–28.16%) and 6.07% (CI 2.83%–9.76%), respectively.

Adding rhinovirus in the best-fitting model I led to the biggest AIC reduction, from 13,216.32 to 13,167.31, when its contribution was quantified by an age-specific coefficient θ . Lastly, the addition of RSV further contributed to AIC reduction (13,153.89). Hence, the final model took the form

$$\mu_{IPD,t,a} = pop_t v_{t,a} + \tau_a Flu_{t-1} + \theta_a rhinovirus_{t-1} + \zeta_a RSV_{t-1} + + \lambda_a IPD_{t-1} + \phi_a \sum_{k \neq a} c_{j,i} IPD_{k \neq a,t-1}$$
(6)

where $v_{t,a} = exp[\alpha_a + \gamma_a \sum_{q=1}^{5} w_q(weather)temp_{t-q} + \delta \sum_{q=1}^{5} w_q(weather)rain_{t-q})]$. As for the model with only Flu, because of large uncertainty about coefficients close to 0, the coefficients $\theta_{5-14}, \theta_{15-44}, \zeta_{5-14}$, and ζ_{15-44} were fixed to zero (models J and K). Fitted values for all age

Age, years	α	γ	δ	$log(\psi)$	$log(\tau)$	$log(\lambda)$	$log(\phi)$
<5	-2.2818	-0.3173	-0.0372	2.5131	-	2.1816	1.2908
5-14	-4.4221	-0.3688	-0.0372	1.5905	-3.2617	2.2835	1.3644
15-44	-4.0347	-0.4770	-0.0372	3.024	-1.6285	3.7285	4.0934
45-64	-2.7756	-0.3419	-0.0372	3.1729	-2.1502	3.4786	3.6104
65+	-1.9382	-0.4640	-0.0372	3.1881	-	3.4232	4.2462

Table 4. Model I: Coefficient estimates for the age-specific model of IPD including Flu.

Abbreviation: IPD, invasive pneumococcal disease.

Age, years	α	γ	δ	$log(\psi)$	$log(\tau)$	$log(\lambda)$	$log(\phi)$
<5	0.0033	0.0017	0.0012	0.0385	-	0.0109	0.0142
5-14	0.0145	0.0092	0.0012	0.0718	0.1035	0.0517	0.0222
15-44	0.0063	0.006	0.0012	0.0220	0.0661	0.0068	9e-04
45-64	0.0128	0.0113	0.0012	0.0158	0.7316	0.0036	0.001
65+	0.0017	0.0011	0.0012	0.0101	-	0.0018	0.0012

Table 5. Model I: Coefficient standard errors for the age-specific model of IPD including Flu.

Abbreviation: IPD, invasive pneumococcal disease.

https://doi.org/10.1371/journal.pmed.1002829.t005

groups are plotted in <u>S7–S11</u> Figs; coefficients and standard errors are listed in <u>S4</u> and <u>S5</u> Tables, whereas the relative contribution of the components is described in <u>Table 6</u>.

Model K showed that the association between RSV and IPD was strongest in the elderly (3.91%, CI 1.83%–6.38%, of cases in the 65+ group and 4.18%, CI 1.58%–6.91% of cases in the 45–64 group), and rhinovirus played an important role in the same age groups: 5.43% (CI 2.23%–8.91%) in the 45–64 group and 5.68% (CI 3.03%–8.32%) in the 65+ group.

Discussion

Using English surveillance data, we quantified the magnitude of the interaction between influenza virus and *S. pnuemoniae* in seasonal and pandemic settings by proposing a multivariate extension of the HHH modelling framework. Such interaction was estimated to be quite small when looking at population-wide counts (model D). These results are consistent with previous research, showing a small association at aggregate level [24]. We found evidence to support the hypothesis of an age-specific interaction [34], the contribution of Flu towards IPD being significant in school-age children and adults aged 15–44 but not in other age groups (model I). Moreover, the components of IPD explained by influenza were strikingly higher during the







2009 pandemic period in the same age groups. This supports findings of Weinberger and colleagues [49]. Other viruses also appeared to interact with *S. pneumoniae* with various intensities across age groups: both RSV and rhinovirus played an important role in 45- to 64- and 65 +-year-olds (models F and K respectively). Such findings support previous evidence of interplay among these pathogens, with differential behaviour across ages [50, 51].



https://doi.org/10.1371/journal.pmed.1002829.g007





The additive structure of the model allowed us to quantify the contribution of multiple viruses to the IPD counts, and at the same time the multivariate age-specific model allowed a better characterisation of each of these interactions. Another important advantage of the modelling framework used here was the potential to assess pneumococcal disease



Endemic	Influenza	Rhinovirus	RSV	IPD Same Age	IPD Other Age
50.35 (34.23-66.91)	0.00	4.49 (<0.01-12.20)	1.31 (<0.01-5.26)	26.32 (16.24-33.95)	17.53 (3.28-32.61)
49.94 (21.31–75.08)	8.54 (4.21-13.43)	0.00	0.00	15.68 (5.06-24.10)	25.84 (1.65-55.97)
26.35 (15.85-38.35)	3.56 (1.69-5.82)	0.00	0.00	19.40 (10.59-26.88)	50.70 (38.19-63.20)
29.24 (20.87–39.76)	0.91 (<0.01-2.83)	5.43 (2.23-8.91)	4.18 (1.58-6.91)	17.15 (8.04–24.27)	43.09 (32.64–53.07)
29.05 (21.65-38.27)	0.00	5.68 (3.03-8.32)	3.91 (1.83-6.38)	23.75 (14.97-30.68)	37.62 (29.41-46.45)
	50.35 (34.23-66.91) 49.94 (21.31-75.08) 26.35 (15.85-38.35) 29.24 (20.87-39.76) 29.05 (21.65-38.27)	Influenza Influenza 50.35 (34.23-66.91) 0.00 49.94 (21.31-75.08) 8.54 (4.21-13.43) 26.35 (15.85-38.35) 3.56 (1.69-5.82) 29.24 (20.87-39.76) 0.91 (<0.01-2.83)	Influenza Influenza Influenza 50.35 (34.23-66.91) 0.00 4.49 (<0.01-12.20)	Influenza Kilitovirus KSV 50.35 (34.23-66.91) 0.00 4.49 (<0.01-12.20)	Indentit Indentit Killiovirus KSV IPD same Age 50.35 (34.23-66.91) 0.00 4.49 (<0.01-12.20)

Table 6. Model K: Relative proportions (%) of IPD cases attributed to pneumococcal transmission within and across age groups, to influenza, rhinovirus, and RSV.

Abbreviations: IPD, invasive pneumococcal disease; RSV, respiratory syncytial virus.

https://doi.org/10.1371/journal.pmed.1002829.t006

transmission. Our findings suggested that 50.70% (CI 38.19%–63.20%) of pneumococcal disease in adults aged 15–44 years, potential parents of young children, was transmitted from other age groups. Transmission within group, on the other hand, prevailed in preschool children and 65+-year-olds: 26.32% (CI 16.24%–33.95%) and 23.75% (CI 14.97%–30.68%), respectively (model K). We speculate this could be due to higher incidence of IPD in care homes or in immunocompromised people.

Finally, the endemic component captured considerable proportions of IPD incidence in all age groups. We can think of this seasonal background as the proportion of disease probably due to some common environmental factors. The adequacy of temperature and rainfall observations to replace harmonic functions, supported by enhanced model fit both at aggregate and age-specific level, reinforces this hypothesis. The appropriateness of shared coefficients for rainfall also suggests that disease seasonality has similar timing across the entire population.

Any estimates of association between two pathogens such as influenza and pneumococcus, both transmitted through air droplets and typical of the winter season, are fraught with uncertainties. First, the validity of this modelling strategy relies on the assumption that viral surveillance is consistent over time and adequately represents the true burden in the population [52]. If this assumption does not hold, apparent trends over time might be due to improved diagnostics or enhanced reporting rather than to a real change in incidence. Our analysis accounted for imperfect detection and reporting of influenza using primary care data and integrating results of virological testing. Reverse transcription PCR (RT-PCR) testing of respiratory specimens is the "gold standard" for confirming a viral presence, and the long-term sentinel scheme implies doctors are not solicited to test because of increased alertness. However, only a subset of infected people seek healthcare, symptomatic disease being a necessary condition. Further, test results may still be subject to misclassification depending on when the specimen is obtained during the illness, and timeliness in reporting results can vary across clinical practices. Future work might exploit the availability of serological testing to better approximate the magnitude and timing of any influenza outbreak. Finally, we simply multiplied the proportion of positive samples by the ILI rates, whereas a joint modelling approach would take uncertainty into account. In terms of IPD data, we believe that testing policies must be consistent over time because of the life-threatening nature of such a condition. Thanks to UK-wide guidelines [40], we believe that reporting was relatively stable over time, making surveillance data as reliable as possible. Nonetheless, the limited numbers of cases, especially in the age-specific analysis, made the resulting estimates uncertain.

Despite our efforts to mimic disease mechanisms, a number of assumptions were made in our analysis. First, we assumed the lag between events to be at least 1 week. Dealing with weekly data, this was the best approximation we could choose within a biologically plausible range [53]. However, the infectious time might be shorter than the chosen time unit, and any unknown delay in reporting might introduce some bias. Second, we assumed autoregressive

coefficients to be fixed over time. This implies that pneumococcal transmission has no seasonal behaviour, and likewise, the interaction with influenza is the same in summer and winter. Such an assumption is meant to include any season-specific variation into the endemic component that summarises a number of unknown aspects such as, for example, climatic influence on disease susceptibility. Fixed autoregressive coefficients also helped to keep our model easy to interpret and to avoid overfitting. Third, contact patterns across age groups were approximated by the POLYMOD matrix, which was estimated on a sample of people living in England in 2005–2006 [48]. Current patterns might be different, and real contact probabilities might not be constant over time. Nevertheless, the use of age-structured contact patterns led to improved model fit compared to an assumption of random mixing between age groups. Fourth, we are aware that pneumococcus is often carried by healthy individuals who might silently transmit the pathogen. In the present analysis, we could not disentangle pneumococcal carriage from disease, as that would have required detailed individual information, such as testing asymptomatic people to detect carriage. Compartmental models with mechanistic assumptions could be employed in future work to fully reconstruct the epidemic process [54].

Despite the above limitations, our modelling strategy successfully improved existing understanding of interaction between multiple pathogens: our estimates are valuable to quantify the possible contribution of influenza to the burden of IPD in a future pandemic of influenza with similar characteristics to the 2009 pandemic, bearing in mind that it was considered relatively mild, compared for example to the 1918 pandemic. The proposed model could be usefully employed by many countries that rely on infectious disease surveillance for informing policy, in terms of both pandemic preparedness and pneumococcal vaccine introduction. Furthermore, we believe our approach could be valuably applied to retrospectively investigate relationships of other notifiable diseases. For example, the contribution of viruses to secondary bacterial infections due to *Staphylococcus aureus* and *Streptococcus pyogenes* requires future investigation to better inform antibiotic prescription policies.

We have clarified the role of the influenza virus on severe pneumococcal infections, in both seasonal and pandemic settings. Although the seasonal contribution does not appear to be relevant, the interaction with pandemic strains resulted significant, particularly in younger age groups. These findings have implications for pandemic preparedness in terms of advising on antibiotic stockpiles, for which currently there is no clear evidence. Finally, a further extension could tackle spatial dynamics if region-specific counts are available, as they would provide a more detailed understanding of spatiotemporal dependencies inherent to the disease and its drivers. However, dynamics of diseases involve processes at different scales of hosts, space, and time, and the attribution of a causal role of one pathogen or another remains a challenging problem [55].

Supporting information

S1 Checklist. GATHER checklist. (PDF)

S1 Appendix. Analysis strategy. (PDF)

S1 Fig. Fitted IPD values all ages. IPD, invasive pneumococcal disease. (TIFF)

S2 Fig. RSV incidence rate. RSV, respiratory syncytial virus. (TIFF)

S3 Fig. Rhinovirus incidence rate. (TIFF)

S4 Fig. Observed counts: 15–44 years old. (TIFF)

S5 Fig. Observed counts: 45–64 years old. (TIFF)

S6 Fig. Observed counts: 65+ years old. (TIFF)

S7 Fig. Model K: Fitted IPD values for infants. IPD, invasive pneumococcal disease. (TIFF)

S8 Fig. Model K: Fitted IPD values for school-age children. IPD, invasive pneumococcal disease.

(TIFF)

S9 Fig. Model K: Fitted IPD values for young adults. IPD, invasive pneumococcal disease. (TIFF)

S10 Fig. Model K: Fitted IPD values for the 45–64 age group. IPD, invasive pneumococcal disease.

(TIFF)

S11 Fig. Model K: Fitted IPD values for the elderly. IPD, invasive pneumococcal disease. (TIFF)

S1 Table. Multivariate model comparison in terms of AIC and one-step-ahead forecast (**log[s(P,x)]).** AIC, Akaike information criterion. (PDF)

S2 Table. Model I: Coefficient estimates for the age-specific model of IPD including Flu. Since Flu coefficients $\tau_{<5}$ and τ_{65+} were very small, we refitted the model fixing them to 0 to make sure the other parameter estimates were not sensitive to such an assumption. IPD, invasive pneumococcal disease.

(PDF)

S3 Table. Model I: Standard error estimates for the age-specific model of IPD including Flu. Uncertainty around coefficients $\tau_{<5}$ and τ_{65+} was not well estimated. IPD, invasive pneumococcal disease.

(PDF)

S4 Table. Model K: Coefficient estimates for the age-specific model of IPD including Flu, rhinovirus, and RSV. IPD, invasive pneumococcal disease; RSV, respiratory syncytial virus.

(PDF)

S5 Table. Model K: Coefficient standard errors for the age-specific model of IPD including Flu, rhinovirus, and RSV. IPD, invasive pneumococcal disease; RSV, respiratory syncytial virus.

(PDF)

Acknowledgments

The authors would like to thank Professor Leonhard Held and Johannes Bracher for the useful feedback on earlier drafts and the participants of the "Pneumonia Coinfection Workshop" at the London School of Hygiene and Tropical Medicine for the helpful discussions. They also acknowledge the contribution of Zahin Amin, Ana Correa, and Praveen SebastianPillai for providing the data, and they would like to thank patients and practices in the RCGP RSC network for allowing their data to be used for surveillance and research.

Author Contributions

- **Conceptualization:** Chiara Chiavenna, Anne M. Presanis, Richard G. Pebody, Daniela De Angelis.
- **Data curation:** Chiara Chiavenna, Andre Charlett, Simon de Lusignan, Shamez Ladhani, Richard G. Pebody.

Formal analysis: Chiara Chiavenna.

Funding acquisition: Richard G. Pebody, Daniela De Angelis.

Investigation: Chiara Chiavenna, Anne M. Presanis, Richard G. Pebody, Daniela De Angelis.

Methodology: Chiara Chiavenna, Daniela De Angelis.

Resources: Simon de Lusignan, Shamez Ladhani.

Software: Chiara Chiavenna.

Supervision: Anne M. Presanis, Richard G. Pebody, Daniela De Angelis.

Validation: Chiara Chiavenna.

Writing - original draft: Chiara Chiavenna.

Writing – review & editing: Chiara Chiavenna, Anne M. Presanis, Andre Charlett, Simon de Lusignan, Shamez Ladhani, Richard G. Pebody, Daniela De Angelis.

References

- 1. National Risk Register of Civil Emergencies—2017 Edition. 2017 Sep. Available from: https://www.gov.uk/government/publications/national-risk-register-of-civil-emergencies-2017-edition.
- Mina MJ, Klugman KP. The role of influenza in the severity and transmission of respiratory bacterial disease. The Lancet Respiratory Medicine. 2014; 2(9):750–763. https://doi.org/10.1016/S2213-2600(14) 70131-6 PMID: 25131494
- Shrestha S, Foxman B, Berus J, Van Panhuis WG, Steiner C, Viboud C, et al. The role of influenza in the epidemiology of pneumonia. Scientific reports. 2015; 5:15314. <u>https://doi.org/10.1038/srep15314</u> PMID: 26486591
- Taubenberger JK, Morens DM. 1918 influenza: the mother of all pandemics. Emerging infectious diseases, 2006; 12(1):15. https://doi.org/10.3201/eid1201.050979 PMID: 16494711
- Morens DM, Taubenberger JK, Fauci AS. Predominant role of bacterial pneumonia as a cause of death in pandemic influenza: implications for pandemic influenza preparedness. Journal of Infectious Diseases. 2008; 198(7):962–970. https://doi.org/10.1086/591708 PMID: 18710327
- 6. McCullers JA. The co-pathogenesis of influenza viruses with bacteria in the lung. Nature Reviews Microbiology. 2014; 12(4):252–262. https://doi.org/10.1038/nrmicro3231 PMID: 24590244
- Davis BM, Aiello AE, Dawid S, Rohani P, Shrestha S, Foxman B. Influenza and community-acquired pneumonia interactions: the impact of order and time of infection on population patterns. American journal of epidemiology. 2012; 175(5):363–367. https://doi.org/10.1093/aje/kwr402 PMID: 22247048
- Chertow DS, Memoli MJ. Bacterial coinfection in influenza: a grand rounds review. Jama. 2013; 309 (3):275–282. https://doi.org/10.1001/jama.2012.194139 PMID: 23321766

- Thornton HV, Blair PS, Lovering AM, Muir P, Hay AD. Clinical presentation and microbiological diagnosis in paediatric respiratory tract infection: a systematic review. Br J Gen Pract. 2015; 65(631):e69–e81. https://doi.org/10.3399/bjgp15X683497 PMID: 25624310
- Imai C, Hashizume M. A systematic review of methodology: time series regression analysis for environmental factors and infectious diseases. Tropical medicine and health. 2015; 43(1):1–9. https://doi.org/10.2149/tmh.2014-21 PMID: 25859149
- 11. Deyle ER, Maher MC, Hernandez RD, Basu S, Sugihara G. Global environmental drivers of influenza. Proceedings of the National Academy of Sciences. 2016; 113(46):13081–13086.
- Trotter Y Jr, Dunn FL, Drachman RH, Henderson DA, Pizzi M, Langmuir AD, et al. Asian influenza in the United States, 1957–1958. American journal of hygiene. 1959; 70(1):34–50. PMID: 13670166
- Jackson M, Peterson D, Nelson J, Greene S, Jacobsen S, Belongia E, et al. Using winter 2009–2010 to assess the accuracy of methods which estimate influenza-related morbidity and mortality. Epidemiology and infection. 2015; 143(11):2399–2407. <u>https://doi.org/10.1017/S0950268814003276</u> PMID: 25496703
- Serfling RE. Methods for current statistical analysis of excess pneumonia-influenza deaths. Public health reports. 1963; 78(6):494. PMID: <u>19316455</u>
- Clifford RE, Smith J, Tillett HE, Wherry PJ. Excess mortality associated with influenza in England and Wales. International journal of epidemiology. 1977; 6(2):115–128. <u>https://doi.org/10.1093/ije/6.2.115</u> PMID: 892976
- Izurieta HS, Thompson WW, Kramarz P, Shay DK, Davis RL, DeStefano F, et al. Influenza and the rates of hospitalization for respiratory disease among infants and young children. New England Journal of Medicine. 2000; 342(4):232–239. https://doi.org/10.1056/NEJM200001273420402 PMID: 10648764
- Thompson WW, Shay DK, Weintraub E, Brammer L, Cox N, Anderson LJ, et al. Mortality associated with influenza and respiratory syncytial virus in the United States. Jama. 2003; 289(2):179–186. PMID: 12517228
- Gay N, Andrews N, Trotter C, Edmunds W. Estimating deaths due to influenza and respiratory syncytial virus-reply. JAMA. 2003; 289(19):2499–2502. <u>https://doi.org/10.1001/jama.289.19.2499-a</u> PMID: 12759316
- Simonsen L, Blackwelder W, Reichert T, Miller M. Estimating deaths due to influenza and respiratory syncytial virus. JAMA. 2003; 289(19):2499–2502. <u>https://doi.org/10.1001/jama.289.19.2499-a</u> PMID: 12759316
- 20. McCullagh P. Generalized linear models. Routledge; 2018.
- Best NG, Ickstadt K, Wolpert RL. Spatial Poisson regression for health and exposure data measured at disparate resolutions. Journal of the American statistical association. 2000; 95(452):1076–1088.
- 22. Cromer D, van Hoek AJ, Jit M, Edmunds WJ, Fleming D, Miller E. The burden of influenza in England by age and clinical risk group: a statistical analysis to inform vaccine policy. Journal of Infection. 2014; 68(4):363–371. https://doi.org/10.1016/j.jinf.2013.11.013 PMID: 24291062
- Matias G, Taylor RJ, Haguinet F, Schuck-Paim C, Lustig RL, Fleming DM. Modelling estimates of agespecific influenza-related hospitalisation and mortality in the United Kingdom. BMC public health. 2016; 16(1):481.
- Kuster SP, Tuite AR, Kwong JC, McGeer A, Fisman DN, Network TIBD, et al. Evaluation of coseasonality of influenza and invasive pneumococcal disease: results from prospective surveillance. PLoS Med. 2011; 8(6):e1001042. https://doi.org/10.1371/journal.pmed.1001042 PMID: 21687693
- Weinberger DM, Harboe ZB, Viboud C, Krause TG, Miller M, Mølbak K, et al. Pneumococcal disease seasonality: incidence, severity and the role of influenza activity. European Respiratory Journal. 2014; 43(3):833–841. https://doi.org/10.1183/09031936.00056813 PMID: 24036243
- Nicoli EJ, Trotter CL, Turner KM, Colijn C, Waight P, Miller E. Influenza and RSV make a modest contribution to invasive pneumococcal disease incidence in the UK. Journal of Infection. 2013; 66(6):512–520. https://doi.org/10.1016/j.jinf.2013.02.007 PMID: 23473714
- Hendriks W, Boshuizen H, Dekkers A, Knol M, Donker GA, van der Ende A, et al. Temporal cross-correlation between influenza-like illnesses and invasive pneumococcal disease in The Netherlands. Influenza and Other Respiratory Viruses. 2017; 11(2):130–137. https://doi.org/10.1111/irv.12442 PMID: 27943624
- Unkel S, Farrington C, Garthwaite PH, Robertson C, Andrews N. Statistical methods for the prospective detection of infectious disease outbreaks: a review. Journal of the Royal Statistical Society: Series A (Statistics in Society). 2012; 175(1):49–82.
- Held L, Höhle M, Hofmann M. A statistical framework for the analysis of multivariate infectious disease surveillance counts. Statistical modelling. 2005; 5(3):187–199.

- Held L, Hofmann M, Höhle M, Schmid V. A two-component model for counts of infectious diseases. Biostatistics. 2006; 7(3):422–437. https://doi.org/10.1093/biostatistics/kxj016 PMID: 16407470
- Paul M, Held L, Toschke AM. Multivariate modelling of infectious disease surveillance data. Statistics in medicine. 2008; 27(29):6250. https://doi.org/10.1002/sim.3440 PMID: 18800337
- Stensballe LG, Hjuler T, Andersen A, Kaltoft M, Ravn H, Aaby P, et al. Hospitalization for respiratory syncytial virus infection and invasive pneumococcal disease in Danish children aged <2 years: a population-based cohort study. Clinical infectious diseases. 2008; 46(8):1165–1171. <u>https://doi.org/10.1086/</u> 529438 PMID: 18444851
- Launes C, de Sevilla MF, Selva L, Garcia-Garcia JJ, Pallares R, Muñoz-Almagro C. Viral coinfection in children less than five years old with invasive pneumococcal disease. The Pediatric infectious disease journal. 2012; 31(6):650–653. https://doi.org/10.1097/INF.0b013e31824f25b0 PMID: 22333697
- Murdoch DR, Jennings LC. Association of respiratory virus activity and environmental factors with the incidence of invasive pneumococcal disease. Journal of Infection. 2009; 58(1):37–46. https://doi.org/ 10.1016/j.jinf.2008.10.011 PMID: 19042025
- **35.** Shaman J, Kohn M. Absolute humidity modulates influenza survival, transmission, and seasonality. Proceedings of the National Academy of Sciences. 2009; 106(9):3243–3248.
- Dowell SF. Seasonal variation in host susceptibility and cycles of certain infectious diseases. Emerging infectious diseases. 2001; 7(3):369. https://doi.org/10.3201/eid0703.010301 PMID: 11384511
- 37. Opatowski L, Varon E, Dupont C, Temime L, van der Werf S, Gutmann L, et al. Assessing pneumococcal meningitis association with viral respiratory infections and antibiotics: insights from statistical and mathematical models. Proceedings of the Royal Society of London B: Biological Sciences. 2013; 280 (1764):20130519.
- Shrestha S, Foxman B, Weinberger DM, Steiner C, Viboud C, Rohani P. Identifying the interaction between influenza and pneumococcal pneumonia using incidence data. Science translational medicine. 2013; 5(191):191ra84–191ra84. https://doi.org/10.1126/scitransImed.3005982 PMID: 23803706
- Correa A, Hinton W, McGovern A, van Vlymen J, Yonova I, Jones S, et al. Royal College of General Practitioners Research and Surveillance Centre (RCGP RSC) sentinel network: a cohort profile. BMJ open. 2016; 6(4):e011092. https://doi.org/10.1136/bmjopen-2016-011092 PMID: 27098827
- Public Health England. Standards for microbiology investigations (SMI). 2014. Available from: https://www.gov.uk/government/collections/standards-for-microbiology-investigations-smi.
- ONS. England population mid-year estimate. Office for National Statistics; 2017. Available from: https://www.ons.gov.uk/peoplepopulationandcommunity/populationandmigration/populationestimates/datasets/populationestimatesforukenglandandwalesscotlandandnorthernireland
- Parker DE, Legg TP, Folland CK. A new daily central England temperature series, 1772–1991. International Journal of Climatology. 1992; 12(4):317–342. https://doi.org/10.1002/joc.3370120402
- Stevens GA, Alkema L, Black RE, Boerma JT, Collins GS, Ezzati M, et al. Guidelines for accurate and transparent health estimates reporting: the GATHER statement. PLoS Med. 2016; 13(6):e1002056.
- Presanis AM, Pebody RG, Birrell PJ, Tom BD, Green HK, Durnall H, et al. Synthesising evidence to estimate pandemic (2009) A/H1N1 influenza severity in 2009–2011. The Annals of Applied Statistics. 2014; 8(4):2378–2403.
- Cox DR, Gudmundsson G, Lindgren G, Bondesson L, Harsaae E, Laake P, et al. Statistical analysis of time series: Some recent developments [with discussion and reply]. Scandinavian Journal of Statistics. 1981; 8(2):93–115.
- Meyer S, Held L, et al. Power-law models for infectious disease spread. The Annals of Applied Statistics. 2014; 8(3):1612–1639.
- Meyer S, Held L. Incorporating social contact data in spatio-temporal models for infectious disease spread. Biostatistics. 2017; 18(2):338–351. https://doi.org/10.1093/biostatistics/kxw051 PMID: 28025182
- Mossong J, Hens N, Jit M, Beutels P, Auranen K, Mikolajczyk R, et al. Social contacts and mixing patterns relevant to the spread of infectious diseases. PLoS Med. 2008; 5(3):e74. <u>https://doi.org/10.1371/journal.pmed.0050074</u> PMID: 18366252
- 49. Weinberger DM, Simonsen L, Jordan R, Steiner C, Miller M, Viboud C. Impact of the 2009 influenza pandemic on pneumococcal pneumonia hospitalizations in the United States. Journal of Infectious Diseases. 2011; 205(3):458–465. https://doi.org/10.1093/infdis/jir749 PMID: 22158564
- 50. Karppinen S, Teräsjärvi J, Auranen K, Schuez-Havupalo L, Siira L, He Q, et al. Acquisition and transmission of Streptococcus pneumoniae are facilitated during rhinovirus infection in families with children. American journal of respiratory and critical care medicine. 2017; 196(9):1172–1180. <u>https://doi.org/10.1164/rccm.201702-0357OC PMID: 28489454</u>

- Weinberger DM, Klugman KP, Steiner CA, Simonsen L, Viboud C. Association between respiratory syncytial virus activity and pneumococcal disease in infants: a time series analysis of US hospitalization data. PLoS Med. 2015; 12(1):e1001776. <u>https://doi.org/10.1371/journal.pmed.1001776</u> PMID: 25562317
- 52. Jackson ML. Confounding by season in ecologic studies of seasonal exposures and outcomes: examples from estimates of mortality due to influenza. Annals of Epidemiology. 2009; 19(10):681–691. https://doi.org/10.1016/j.annepidem.2009.06.009 PMID: 19700344
- McCullers JA. Insights into the interaction between influenza virus and pneumococcus. Clinical microbiology reviews. 2006; 19(3):571–582. https://doi.org/10.1128/CMR.00058-05 PMID: 16847087
- Opatowski L, Baguelin M, Eggo RM. Influenza interaction with cocirculating pathogens and its impact on surveillance, pathogenesis, and epidemic profile: A key role for mathematical modelling. PLoS Pathog. 2018; 14(2):e1006770. https://doi.org/10.1371/journal.ppat.1006770 PMID: 29447284
- Gog JR, Pellis L, Wood JL, McLean AR, Arinaminpathy N, Lloyd-Smith JO. Seven challenges in modeling pathogen dynamics within-host and across scales. Epidemics. 2015; 10:45–48. <u>https://doi.org/10.1016/j.epidem.2014.09.009</u> PMID: 25843382